Author's response to reviews

Title: Utilization of delivery care for rural women in China: does the New Cooperative Medical Scheme (NCMS) make a difference? A population-based cross-sectional study

Authors:

Shengbin Xiao (tixsb@mail.xjtu.edu.cn)
Hong Yan (xjtu_yh.paper@yahoo.com.cn)
Yuan Shen (shenyuan_xjd@163.com)
Shaonong Dang (tijdshn@mail.xjtu.edu.cn)
Elina Hemminki (elina.hemminki@thl.fi)
Duolao Wang (duolao.wang@lshtm.ac.uk)
Qian Long (qian.long@thl.fi)
Jianmin Gao (gaojm@mail.xjtu.edu.cn)

Version: 4 Date: 29 October 2010

Author's response to reviews: see over
Title: Utilization of delivery care among rural women in China: does the Health Insurance make a difference? A cross-sectional study

Authors: Shengbin Xiao, Hong Yan, Yuan Shen, Shaonong Dang, Elina Hemminki, Duolao Wang, Qian Long and Jianmin Gao.

Dear Editor,

We have now checked and corrected the manuscript (MS: 2051661291426391) according to the requests. Some typing errors had been found. What has been changed in the revised manuscript are as follows:

1. Highlighting and tracking has been removed from the manuscript.
2. 'April' has been changed to 'April 2008' in line4 in Method section.
3. 'area' has been changed to 'areas' in line2 in Conclusions section.
4. 'Method' has been changed to 'Methods' (subtitle) in page 4.
5. Figure1 has been updated for improving the image effects.

Thank you!